' 1rz KfeViCourse Overviewuuing Users andTheir Tasksuples and Guidung with suon StylesuUI Design EleuuUI Deveent ToolsuIterative Dn andility Testiuct Presentations acted TopicsuCase unt Devents inu ' 1rz KfeViChapter OverviewVisual Design Guidelinesuuun Layout and DuGeneral LuuuCoding uuations for DesignuuWeb Punt Concepts andTermsupter Summ ' 1rz KfeViSlides FeedbackuI would like to gbout thel desigint Slides I�m using. Pleawrite down your imprs (positive or negative) on thewing aspects:vGeneral l-v-italicsv-vCoding -. ' 1rz KfeViImportant UI Design ElementsuWhich of for scuuu ' 1rz KfeViMotivationuMost of the visthe screen, and it is very .uThe scruGood scrhe effort rthe user to solve a task, e.g. by over recall, by providing clinputs, or by sother attri ' 1rz KfeViufor scudesign for specific suto effectively uanslate task or user re scu

' 1rz KfeViEvaluation Criteriaurelevant viurecall of the most ufor specific uidual screens for specific ,creens fasks ' 1rz KfeViVisual Design ' 1rz KfeViScreen Layout and Designuprinciples of scto any kof interfus, fill-in forms, displays of database Web paut work onstraints tware of a sun sizevr of pixels, pixel tion, screen oon (horizontal orulay capvg, reverse video, bit-mapped g, color, sou[Mo] ' 1rz KfeVi0Screen Layout Categoriesufive ural lauuung technsur[Mo] ' 1rz KfeVi1General Layout Principlesuscope of uuu[Mo] ' 1rz KfeVi2Scope of informationuential to decisud putting too much tion on the screenur put more on the screen that absolutely necube brief and coeful to match the level ofl to the user�s knoe and ve.g. diying on-screen ructions . type of user (expert v[Mo]

' 1rz KfeVi3Scope of Information (cont.)uential to decisueach screen should stand avining one idea or one tuusers should never have tovanother screen to accsh a task on the current screenver information on one screen to ete anouerror ms should be co, and not requirence to manuals to her or unvpoor: �file name too long�v �file name should not exceed 8 ters� ' 1rz KfeVi4Arrangement of Itemsustart corner of ucking studies show thaer-left corner of a diuish lane is wrihtvmany other languages do notveye also tends to move in a c-wise muthe human brain tendrder and patterns in alay, and follow thre appvt princ-ity---ity-[Mo] ' 1rz KfeVi5Arrangement of Items (cont.)uukeep group size manvroughly 12-14 cers wide by 6-7 lines high yy 5 deg. of visual umize the nr of groupsvr of groups ts search time, even when group size isluprovide symmetry uh the use of blank spacesvbut don�t imbed blank spaces to stretch out a word-�ur words if approp[Mo] ' 1rz KfeVi6Formatting Standardsutently in all screens within a syud screens in a system should always have standtypes of information located in the same pve.g. screen t, status mes, input fields, command , and ages should ly appear in tsame area of the screen throughout the uugn one or more screen tes that follow rules ofn layoutuw the temp[Mo] ' 1rz KfeVi7Fonts and Stylesuutwo dier screen is plentyuok good on paper may not nely lookgood when ayed, and vice veuiewed from a distance, avoid the use ofvuse fonts like or Arialue styustyles are changes within a fontv, , and

vthey also ide different character s-, 16 point, , 20 poi, 24 po[Mo] ' 1rz KfeVi8CAPITALSuutext in all uppercase is more did than text inmixed ves have shown that reading mixed text is about 13% fathan reading upe textuuse unate titles, main cauder the us insteadvif av[Mo]

' 1rz KfeVi9Captions and Fieldsudsua fill-in or display field should be easily dfrom its caption, label, or titleuder the use of contrasting feake capand fields stand outvr, case, inten, font, or reverse [Mo] ' 1rz KfeVi0

QUARTER SPRING 96CE SOR ALLSESERORDATAS03 1SOScreen Design Examples[Mo] ' 1rz KfeVi1

Screen Design Examples[Mo]TER: 02:SESEECOMP 535 Data S 3PSYCH uction to 3S. AdamsBIOL 201tics 3A. PapeCOMP 675 3P. Mustillo Interface DesiPress AR <RETURN> to go back ' 1rz KfeVi2

Screen Design Examples[Mo]COUERINGSQRTER YSPRI96AUENCE LLCOURCOURNUMRNUMBERNAMECREDITSPROCOMP5DATASTRUCITHPSYC1INTROADAMSO1NETICSPAPERMCOMP5HCIMUI SE OFFERINGSQUTER:INERY:LQUENCE:ER:LURRSENUMBERONUMBERNEEDSCOM5DucmPSYC1IsPOACOM6H-CPuso I

to scls <RETURN> to go back ' 1rz KfeVi3Text Layoutuuctions, couuation frothe useruuct a respoe useruutell the user whully longer than messages or pumay be uoften used in help or dntation [Mo] ' 1rz KfeVi4ulabels suuif you can, display the entire wouo the left uleft justify labels iuued labels produce a d margvhard to ut fields [Mo]

' 1rz KfeVi5Example Labels[Mo]

x ' 1rz KfeVi6uof messubrief and conuve; avoid negative mesuctive, not criticalufic and cour is in coutent with a ru(a) poding gear is not down�(b) beding gear is u�put the lng gear dowuof detail acc[Mo] ' 1rz KfeVi7Examples Messages[Mo]PrBett

sng of thxt

ed 23 o u O u t 2 g IRT ZeHIRE r a i 1 4 Cnot efore saving ve f o re e x i dava ld name ile n a m e le h i 8 c h a t n or 1542Unmc tpa h is 210

' 1rz KfeVi8Promptsuun and where they will be neeur prompts cally or seuformat prompts using blank spaces, justification, and visucues for easy scauof detail acc[Mo] ' 1rz KfeVi9Prompts (cont.)uuuse the active tenvsay �Push the buttonot �The button is to be d by the user�use promtive statemvd negati�Do x� vs. �Do not do y�uuse cot terminovuse the word �type� when iting a user to type svuse �Press� for pressing a button or function -ves on labels ld be e--s[Mo] ' 1rz KfeVi0Examples PromptsPrBeer

sion csor and press uPoson c P s s r e n to a c c S___ (8)

messisent by sS n d m s a g P s s T RDot return to menu omping eyCom te en bef o ret u g t m e ge forward after en

adssEer ad d s, t p a ge fo r w s backtab to g, go down go up,Pr s do d o w n, Pr e T ss or upH

n Pwn

' 1rz KfeVi1uukeep raphs shortu separate them bne empty lineulimit lines to 50-55 characters per linevor use double ns, 30-35 characters per coluud right jun with ual spauspace en lines should be equal to or slightly greaterthan character htv. double spacud line ln 26 chaTHAT THEYLOOK LIKECOLUMNS[Mo] ' 1rz KfeVi2Instructions (cont.)uukeep text simple and cluuse short sentences, and e and familiar wud hyphnud acronymsvs you are sure that all users will tand them[Mo] ' 1rz KfeVi3Layout of Numbersuright justify uuuuuse standard stors when they apuise, use s[Mo] ' 1rz KfeVi4Examples Numbers[Mo]PrBeer

10100010

0101000 1 0 0 100256 344841 2 5 6 3 5 , . 4 10:1 p.029610:01 p 7

0021000132100 1351 83880456 (514) 80 8 6 7 61586025615 860 2 ' 1rz KfeVi5Coding Techniquesuuknow your screen device anle ofuuse these techns sparinuking, reverse video, colu, shadow

, embosuCAPITALS and Smu, shape, special crs and iumity, borders, depthud, animation and motion[Mo] ' 1rz KfeVi6Coding Techniques (cont.)uuuse to get the user�s attention. example: critical errorud blin5 Hz, with lr �on� pethan �off� periodvmum �on� period of 50%uudark fornd on light band or light foreground dark bumay reduce lelity, so should not be overuvery powerful search cumay be used to im has been selected orto indicate an error field[Mo]

' 1rz KfeVi7Coding Techniques (cont.)ueucan be used aelative tity or impoulimit to 5 or less levels, because users have difficultying more than fiuucan be used to draw attevase c, doubled characters, cs, and othernt fontsulimit to 2 t character types per screenusame limit applies to the use of different font sizes ' 1rz KfeVi8Coding Techniques (cont.)uuuse to make things stand oid using more three different levels of bossvusers don�t remember the diffuucan be used to make an t or character stand ocially key wordsumay reduce leuld be used sp ' 1rz KfeVi9Coding Techniques (cont.)uumay be used to impart different conns, or impsome mgvred octagon shape in north America is usually ed with suspecial characters utags such as asterisks, arrows, and other gralls, such as a small png hand can be used todraw attention the screen ' 1rz KfeVi0Coding Techniques (cont.)uyumity implies assn, and leads the eye toder itemuuvery effective in idg meal groups, creating asense of closure, and focusing ' 1rz KfeVi1Coding Techniques (cont.)uuvery effective in dg attentionunt sounds can convey closs, magnitude, weuld be ue when used in noisy euunot much known about the use of animation and motion a screen design tech, although it would seegood g or alert mechavthe human eye is very sive to n in the peripheve.g. insect on the ' 1rz KfeVi2Coding Techniques (cont.)uudepth is a powerful cue in navular depth cues include: (a), size, (b) inton, (c)st, clarity, and brig, (d) shadow, (e) texun parallax is a ar powerful depth cuever moving relative to the enent in a train wag thede worlduology nable of rendering true depth on a display, ah significant progress has beenvareas of wire-frame ing, sject mng, and vsome lays now support sopic g

' 1rz KfeVi3Coding Techniques (cont.)uuvery powerful and salient coding teue that can beused in desig screensvto draw anvto cocate orgvto indicate vto esh relativto express vas a search ' 1rz KfeVi4Color (cont.)umust be chosen and used with great carevthere are cal as well as cultural es in how weve cvwe perceive colors tly, dg on age, cognes, ere to color (adaptavr perception is ect to tual e, in which atrs influence one another. for ex, a gray square on a darkground appears lighter and brighter than the same color on alight vr perception is ect to tive s--- ' 1rz KfeVi5Colors and Cultures[Mo]UCA(1ssr asFact2% 61.4e 96.1noww 81.1 99.2 50.4 94.5Danger 89.8 31.5

' 1rz KfeVi6Colorvuse color spayugn first in mvize other aspects of the screen layout nvadd color only where it adds evdon�t use color just because you have a color mouuse color to draw attentiote organn,to indicate statutablish suuse color to support search tasksvr is a more salient cue for search tasks and sltion tasks than other cues, such as shape, vple: cg the number of red ts vs. the number ofntal bars, of small ts, or bright obud using color in non-ted waysuw users to control color co[Mo] ' 1rz KfeVi7Color (cont.)uunot all users may have color didut coding s the inteported to achrome tion without losing impules and 0.4% of females have some sortof color deficieut coding nces pece.ve.g. It is faster, easier, and less prone to error to search fsquare than to rch for objects that are just red or just squ[Mo] ' 1rz KfeVi8Color (cont.)uolors differ in lias well utness reon of dull to brightuhue is the visual sensation that varies ang tothumost common types of color deficievthe inty to dminate red from green and blue from owuall sighted pe can diate differences in lighvness refers to the range of achromatic rs ranging fromwhite through gray to blus may not be clear for colors on a colored bvrs must contrast in ss as well as hueutness is mted by red and green phvrs differing only in blue will not produce sharp e[Mo]

' 1rz KfeVi9Color (cont.)ube consistent with color associations that users in their juin Western cultures, the follg assocs are typicrced in traffic signals and consumer products.vGo, on, v, hot, danger, on, yvon, warning, warmvCold, uTip: Usev for error mesv for warning mes[Mo] ' 1rz KfeVi0Color (cont.)ulimit tinct uably less than fourumake them as distinctive as possived as far away as posle on the visible color suIf y8 colors, duto assist users in lng and ering color coutext linesul acuity is very poor for detailed, sharp, shouhe human eye is not good at focusing on bluevthe fovea (central area of the eye) is ve to bunsitivity to blues ies with authe lens becomes yellow with age[Mo] ' 1rz KfeVi1Color (cont.)uchoose syu display color images on an atic bacd (blagray, or white), or atic images on a ves color contrast and color discure greatest amoust between foregand bund under different lighting va light background could result in increased er and guuse dark backds for coloht backgvy when ing large graphical ss[Mo] ' 1rz KfeVi2Color (cont.)uconsider vbol colorsuing angle and distance influence color puends the vd red and green beyond 40 deg. of ng anglevd yellow beyond 50 deg. of vg anglevd blue beyond 60 deg. of viewing anvuse highly ted colors and colors with high lue confor viewing dies beyond 60 cmvd saturated colors on dark bacds for vg distd 60 cm[Mo] ' 1rz KfeVi3Color (cont.)ucolor sity ureds and blues are more difficult to distinh in dimting than greens and yesvthe eye is most tive to middle wavve.g. a red rose seen at dusk s blackuolder users els toh colorsuthe overall sensitivity oshes with ageuthe lens becomes less transpa[Mo] ' 1rz KfeVi4Color (cont.)uuthe human eye is insensitive to red and green in thevy when these colors are highly saturated, and when thelayed targets or shapes are svtion refers to the number of different wavelengths nt orbuting to the sion of uplays, use yellud the simuous display of highly s,lly extreme colorsve.g. red vs. blue, or w vs. vrequires tant re-fo[Mo]

' 1rz KfeVi5Color (cont.)uuularly fouthe eye oth colors simuvthe use of both colors results in a perceptual non knas csv i.e., red and blue text appears to lie in different depth pl[Mo] ' 1rz KfeVi6Chromostereoscopic Effect[Mo]lie in a different lie in a different lie in a different lie in a different lie in a different lie in a different ' 1rz KfeVi7Summary Screen Layout and DesignThe format and content of innlayed on the screen is veryrtant in deing the success ofa user‘s interaction wiThree ar in screen des¥on Presented (sity)¥ntation on (al Rel)¥lexity on Presen

' 1rz KfeVi8Summary Screen Layout and Designuumize the ation by png onlyt is absolutely nevise wording, far data fovdance of unnecary detauspatial ruping similar items in a dy for readvcan hght relatps between different groups of -uy of ukeep the layoeens as simple as possuies equally for all types of screend interfavuding Web pa[Mo] ' 1rz KfeVi9Implications for Windows Designuprinciples us of[Mo] ' 1rz KfeVi0Buttonsuuse sufficiently uas the size of a button increases, so does the serse to and seleuas the distance of the mouse movement des, speedalso ivknown as Fitt�s Lawvkeep buttons large and mouse movement nce suze consiumake all of the buttons in a og box the same svs you have a very good reason for different

' 1rz KfeVi1Buttons (cont.)uuar functiouto leave the window (OK, Canceuctive actions (Deule capitalizutalize the each word of the button lauyu clear and concise button labuve buttons to leave a window (OK)uling actions to leave the window (Cauugray out or (dim) a button if it is unavail[Mo] ' 1rz KfeVi2Menusuistent ruuse ellipsis (�) in items to uOpen �uumming them outuuse check uto selecattributes [Mo] ' 1rz KfeVi3Menus (cont.)uof casuthey add compeuthey hide choices by ag a layer to menusuare phly more duld only be used to minimize screen real estateuogical relatithey cu[Mo] ' 1rz KfeVi4Implications for Web Designuuuuser interface cons[Mo] ' 1rz KfeVi5Importance of the Webuuone new Web user every 2 seconds (Forrester Re)umore than 150 n Web users woe todayuuuuntvality nt is a major driver for attracting repeat ureach uu[Mo] ' 1rz KfeVi6uWeb sites or browsers suactive table of coucking and history listsusite maumarks (N) or favorites (Internet Explorer)uvdon�t date inn unless it is updated freq.vusers will iately e a site if they see that the information not cvtell users how often tion will be uvupdate tion au[Mo]

' 1rz KfeVi7Design Elementsuframesug a page into frames can be confusing for usersuframes may break the fundl modevts become more comvty of user actions may be viouunot all Web ders are good graphic deuusers have a short attention spanvif downg a page takes er than 10 , they may luolling textudo not include page elemve all ovlming to many users, annoying to othersvusers are very e to motion in the pe-[Mo] ' 1rz KfeVi8Design Elements (cont.)uLsue URLuld contain a hule directory and file navt the nature of the information svevery page should have a link to the home page and an ation of where it will fit within the structure of the informeusumake sure that all pages include a clear indn of whatWeb site they belong tovusers may s pages without going through the home pageuuation oan get ory quick[Mo] ' 1rz KfeVi9uuusers don�t know as much about your site as you douusers gve a site mapv let your users know where they are and where they can goude a good search mechvtimes even the best navigation anism is not enoughvsome users want to jump directly to the page they needuulinks to ave not seen are typically blueulinks to usly visited pages are y purplenta) or redvbe cotent; don�t mess around with color s[Mo] ' 1rz KfeVi0Textuusers scan tuing speeds are >25% slower from computer scruuse scale textvighted words and desce linksvngful svted livone idea per pavrted pyramid style - present cusion vcut down on the number of wordsuusers don�t like to scuonly 10% of users scroll beyond the inle on the screen when a page comes upuall critical inn should be on the top p[Mo] ' 1rz KfeVi1utry our users unew or returning visitorusing, seeking s, downlct?u label all linked gcsu do not leave orphan links, where users must back upvs provide at least a way to get back to the home pageukeep links upvk to see if all links are still vaularly check your links in your dovmake sure they go to the right placeuard text link colors and hi[Mo] ' 1rz KfeVi2Navigation Supportuur and intuitive site infovtable of contents, , site map, �What�s uable and obvious visual ntion schve.g. buttons, text, site muh for topics, keywords, nameutopics ks, and names inarch enude navnal aids on every pagevmake it pose for users to jump right to any page from a Wh engineu let users know where a ation link will take them[Mo]

' 1rz KfeVi3Tasksuperform tasks quicuce searcload pages and imaing within a pulimit number of choices at a levelup important choices togeunize inn and pages to ate frequtasksude an out use blee technology t a good ve.g. use VRML only if you have information that maps ly to3D sp[Mo] ' 1rz KfeVi4Contentukeep content sivs only one topic per pageuontent qve important content at the top of Web y and at the topof the Web paude printable v[Mo] ' 1rz KfeVi5Consistencyuh user interface consiurate identityvs, i, colors, fonts, and vuse the corporate identity cy throughout the Web eustent link and naon schemes throWeb inustent heas and terulay bacd infovURL, cts, e-mail add, etc.ugn the ine easily accessible and usavnt browsers, platforms, screen [Mo] ' 1rz KfeVi6Web ResourcesuvuvuvuvuJakob Nielsvuv[Mo] ' 1rz KfeVi7Post-Testulayout for text, elements likeuList ribe fivcreen layuhe regular scr ' 1rz KfeVi8uuportant taskic factors to consifor the screen lay

' 1rz KfeVi9Important Concepts and Termsuuuuun layutext uuuser interface uual diu

What

You

SeeIsWhat

GIWYG)uWeb duusung tecuruuuual user interfauhuman factors uuoutput dusus ' 1rz KfeVi0Chapter Summaryuthe vistransfer capacity to user, very uual aspects have a stuser interfu, textthe visu ' 1rz KfeVi1